Author's response to reviews

Title: Lung ultrasound: a useful tool in diagnosis and management of bronchiolitis

Authors:

Vincenzo Basile (vinbasile67@libero.it)
Antonio Di Mauro (dimauroantonio@msn.com)
Egisto Scalini (e.scalini@libero.it)
Paolo Comes (paolocomes58@gmail.com)
Ignazio Lofù (i.lofu@tin.it)
Michael Mostert (studiomostert@virgilio.it)
Silvio Tafuri (silvio.tafuri@gmail.com)
Mariano Manzionna (mariano.manzionna@alice.it)

Version: 4 Date: 4 May 2015

Author's response to reviews: see over
Monopoli, 2nd May 2015

To Editor
BMC Pediatrics

Please find enclosed the revised manuscript “Lung ultrasound: a useful tool in diagnosis and management of bronchiolitis” on the basis of the comments we received.

This work has been recently presented during the PAS Meeting – San Diego 2015 (PUBBLICATION NUMBER 4182.625).

Our “Authors Contributions” section has been modified according your format and our ethics statement has been updated by includeing the name of the Institutional Review Board that approved our study, as requested. We hope it’s now suitable for publication in the journal.

We look forward to hearing from you in due course.

Yours sincerely,
Mariano Manzionna

Address for correspondence:

Mariano Manzionna, MD
“S. Giacomo” Hospital, ASL BA, Monopoli
Largo Simone Veneziani, 21, Monopoli (Bari), Italy
mariano.manzionna@alice.it
+39.080.41.49.52.34
+39.380.71.17.355